Author’s response to reviews

Title: Informal Care Relationships and Residential Aged Care Recommendations: Evidence from Administrative Data

Authors:

JEROMEY TEMPLE (jeromey.temple@unimelb.edu.au)

Marijan Jukic (mjukic@student.unimelb.edu.au)

Briony Dow (bdow@unimelb.edu.au)

Version: 2 Date: 26 Oct 2017

Author’s response to reviews:

1. We have noted that you have stated in the Ethics approval and consent to participate section that "These data are collected by the Australian Federal Government under the Aged Care Act and provided to registered users for research purposes." Please clarify whether this study was given permission/was registered to use this data, as this needs to be clearly stated.

Clarified as requested.

2. Please change the Introduction heading to Background.

Changed as requested.

3. Please change the Ethics approval heading to Ethics approval and consent to participate.

Changed as requested.

4. Please clarify whether the patient data used in this study was de-identified, or where the patients aware and consented to their data being used in this study.

Clarified as requested.

5. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:
http://www.biomedcentral.com/submissions/editorial-policies#authorship

Clarified as requested.

6. Please remove any supplementary files which you do not want published, as they are no longer needed at this stage.

OK

7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

OK